Author's response to reviews

Title: The value of functional data analysis (FDA): a systematic review of its application

Authors:

Shahid Ullah (shahid.ullah@flinders.edu.au)
Caroline F Finch (caroline.finch@monash.edu)

Version: 3 Date: 12 December 2012

Author's response to reviews: see over
12 December 2012

Adrian Aldcroft
Executive Editor
BMC Medical Research Methodology

Re. Resubmission of manuscript “The value of functional data analysis (FDA): a systematic review of its application”

Dear Adrian,

I would like to re-submit our paper entitled “The value of functional data analysis (FDA): a systematic review of its application” to BMC Medical Research Methodology for your consideration. The authors responded the Reviewer’s question in details. The manuscript has been extensively checked for grammar and English language expression. Where necessary, edits have been made to the text to improve its readability.

The co-author and I certify that we have both participated sufficiently in the work to attest to its overall validity and to take public responsibility for appropriate portions of its content. I hope the above information will merit your enthusiastic response, and I look forward to hearing from you. I would be happy to provide you with any other information you may require.

Endorse: Manuscripts

Thanking you.

Yours sincerely,

Shahid Ullah, PhD
The value of functional data analysis (FDA): a systematic review of its application

Shahid Ullah¹* and Caroline F Finch²

¹Flinders Centre for Epidemiology and Biostatistics, School of Medicine, Faculty of Health Sciences, Flinders University, SA 5001, Australia (shahid.ullah@flinders.edu.au)

²Australian Centre for Research into Injury in Sport and its Prevention (ACRISP), Monash Injury Research Institute (MIRI), Monash University, VIC 3800, Australia (caroline.finch@monash.edu)

*Address for correspondence:

Dr Shahid Ullah, Flinders Centre for Epidemiology and Biostatistics, School of Medicine, Faculty of Health Sciences, Discipline of General Practices, Health Sciences Building, Flinders University, SA 5001, Australia. Ph: +61 8 7221 8518, Fax: +61 8 7221 8544. Email: shahid.ullah@flinders.edu.au